Author's response to reviews

Title: Patient safety in primary care: A survey of general practitioners in the Netherlands

Authors:

Sander Gaal (s.gaal@iq.umcn.nl)
Wim Verstappen (w.verstappen@iq.umcn.nl)
Michel Wensing (m.wensing@iq.umcn.nl)

Version: 4 Date: 10 December 2009

Author's response to reviews: see over
Revision note

Thank you for your response.

The ethical committee of the Radboud University Nijmegen Medical Centre approved the study. The declaration is added below.

We added the following statement on page 4: “The ethical committee of the Radboud University Nijmegen Medical Centre approved the study.”

With kind regards,

Sander Gaal
Commissie Mensgebonden Onderzoek
Regio Arnhem – Nijmegen

Cannisius-Willemius Ziekenhuis
UMC St Radboud
Ziekenhuis Rijnstate AlyesZorggroep
St. Maartenkliniek

Drs. H.A. Gaal,
IQ healthcare,
Geert Grootplein noord 21,
6525 EZ Nijmegen

Ons Kenmerk
CD/CMO 2470

Datum
7th of December 2009

Titel: Patiënt safety in primary care: A survey of general practitioners in the Netherlands
CMO dossiernr: 2009/324

Dear mr Gaal,

On behalf of the research ethics committee of the Radboud University Nijmegen Medical Centre I hereby let you know that the abovementioned study has been carried out in the Netherlands in accordance with the applicable rules concerning the conduct of medical research and the review of research ethics committees.

Best regards,
On behalf of the CMO Regio Arnhem-Nijmegen
Dr F. Huysmans, Chairman

Postadres
CMO Regio Arnhem-Nijmegen
p/a UMC St Radboud
Huispost / Route 578
Postbus 9101
6500 HB Nijmegen

Bezoekadres
UMC St Radboud
Th. Crooniuslaan 6
6525 GH Nijmegen

Tel. 024-3613154
Fax 024-3648148
E-mail cmr@medradku.umcn.nl
Internet www.umcn.nl